Author’s response to reviews

Title: Do Pregnancies Reduce Iron Overload in HFE Hemochromatosis Women? Results from an Observational Prospective Study

Authors:

Virginie Scotet (virginie.scotet@inserm.fr)
Philippe Saliou (philippe.saliou@chu-brest.fr)
Marianne Uguen (virginie.scotet@gmail.com)
Marie-Christine Mérour (marie-christine.merour@efs.sante.fr)
Céline Tripogney (celine.triponey@efs.sante.fr)
Brigitte Chanu (brigitte.chanu@efs.sante.fr)
Jean-Baptiste Nousbaum (jean-baptiste.nousbaum@chu-brest.fr)
Gérald Le Gac (gerald.legac@chu-brest.fr)
Claude Férec (claude.ferec@univ-brest.fr)

Version: 2 Date: 30 Nov 2017

Author’s response to reviews:

Dear Editor,

Please find enclosed a revised version of our manuscript entitled “Do Pregnancies Reduce Iron Overload in HFE Hemochromatosis Women? Results from an Observational Prospective Study” and referenced PRCH-D-16-00615_R2, as well as a point-by-point responses to the editor and reviewers comments.

Hoping that our manuscript will now be accepted for publication.

Sincerely yours,

Claude Férec & Virginie Scotet
POINT-BY-POINT RESPONSES TO THE REVIEWERS

Editor Comments:

Please move the ethics approval and consent to participate information to the Declarations section under appropriate headers. Please also check though the manuscript for English language and grammar. Thank you.

As requested by the Editor, we moved the paragraph on ethical issues to the Declarations section under a heading entitled “Ethics approval and consent to participate”. We also sent the manuscript to a native speaker for English correction/editing.

Reviewer reports:

Kyly Whitfield (Reviewer 1): Thank you to the authors for their thorough edit of their manuscript. Paper looks terrific, and will be a great addition to the journal, and the literature at large.

Eva K. Pressman (Reviewer 2): All of my concerns have been addressed.